Author's response to reviews

Title: Development of a complex intervention to test the effectiveness of peer support in type 2 diabetes

Authors:

Gillian Paul Ms (gpaul@tcd.ie)
Tom O'Dowd Prof (todowd@tcd.ie)
David Whitford Prof (dwhitford@rcsi.ie)
Fergus O'Kelly Prof (feokelly@tcd.ie)
Susan M Smith Dr (susmith@tcd.ie)

Version: 4 Date: 19 July 2007

Author's response to reviews: see over
19\textsuperscript{th} July 2007

Dear Editor,
Thank you for accepting this paper for publication. We have now made all the revisions and corrections outlined in your email of 18\textsuperscript{th} July 2007. We hope these are satisfactory.
Many thanks,
Ms Gillian Paul (on behalf of co-authors)